CSC108H lab – week 12

This document contains the instructions for the week 12 CSC108H lab. To earn your lab mark, you must actively participate in the lab.

In this lab, you will verify your A4 partner’s information, and write a few methods to insert and remove items from an array.

1 Sorted arrays

Take out your lecture notes and turn to the arrays section. You’ll be working with int[]s. Get this file:

http://www.cs.toronto.edu/~heap/108/labs/lab-week12/ArrayStuff.java

It has a main method that tests the various methods. It compiles and runs, but the methods are only stubs. Fill in the bodies in the order they appear.

After each one, compile and test it, and then alternate navigator and driver.